Author's response to reviews

Title: MRI-negative PET-positive Temporal Lobe Epilepsy (TLE) and Mesial TLE differ with Quantitative MRI and PET: a case control study

Authors:

Ross P Carne (carnero@svhm.org.au)
Terence J O'Brien (obrientj@unimelb.edu.au)
Christine J Kilpatrick (Christine.Kilpatrick@mh.org.au)
Lachlan R MacGregor (Lachlan.Macgregor@mh.org.au)
Lucas Litewka (Lucas.Litewka@svhm.org.au)
Rodney J Hicks (Rod.Hicks@petermac.org)
Mark J Cook (Mark.COOK@svhm.org.au)

Version: 4 Date: 16 June 2007

Author's response to reviews: see over
16/06/07

Dear BioMed Central Editorial Production Team,

Re: Authors: Ross P Carne, Terence J O'Brien, Christine J Kilpatrick, Lachlan R MacGregor, Lucas Litewka, Rodney J Hicks and Mark J Cook
Title: MRI-negative PET-positive temporal lobe epilepsy (TLE) and mesial TLE differ with quantitative MRI and PET: a case control study
Journal: BMC Neurology
MS: 1163920501258368

We were delighted to hear that BMC Neurology will publish our manuscript, and have made the following changes to the format of the manuscript as requested.

Formatting changes requested
---------------------------------
Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

Amended as requested.

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system

Amended as requested.

Highlighting/Tracking - Please remove any highlighting/tracking from your manuscript text.

Removed as requested.

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Amended as requested.

Please do not use full capitals for country names.

Amended as requested.

Keywords, Word-counts, Running head and Running title - Please remove these.

Removed as requested.

Remove duplication of Corresponding author e-mail details.
Amended as requested.

Subjects and Methods -> Methods.

Amended as requested.

Insert Competing interests header.

Inserted as requested.

Authors' contributions - Please indicate that all authors read and approved the final manuscript.

Inserted as requested.

References - The reference list should contain all authors’ names, if there are 30 or below. The term ‘et al.’ should not be used.

Amended as requested.

Reference list -> References.

Amended as requested.

Insert Figure legends header.

Inserted as requested.

Order of tables -> Tables should be listed in the order that they are referred to in the text.

Checked and appropriate.

Figures -> Figures must be uploaded in the order in which they are referred to in the manuscript, as they are automatically numbered when they are uploaded.

Checked and correct.

Please let us know if there are any further modifications required.

Thank you again for your trouble and we look forward to working together again in future.

Yours Sincerely,

A/Prof Ross Carne for the Authors.